Author's response to reviews

Title: Home exposure to Arabian incense (bakhour) and asthma symptoms in children: A community survey in two regions in Oman

Authors:

Omar A Al-Rawas (orawas@squ.edu.om)
Abdullah A Al-Maniri (almaniri@gmail.com)
Bazdawi M Al-Riyami (bazdawi@squ.edu.om)

Version: 8 Date: 15 May 2009

Author's response to reviews:

Five minor typo errors were identified and corrected as shown in Track Changes of the resubmitted final version.